Author’s response to reviews

Title: Prevalence of adjustment disorder among cancer patients, and the reach, effectiveness, costutility and budget impact of tailored psychological treatment: study protocol of a randomized controlled trial

Authors:

F.E. Beek (florie.van.beek@vu.nl)
L.M.A. Wijnhoven (Lonneke.Wijnhoven@radboudumc.nl)
F. Jansen (f.jansen1@amsterdamumc.nl)
J.A.E. Custers (Jose.Custers@radboudumc.nl)
E.J. Aukema (e.aukema@olvg.nl)
V.M.H. Coupé (V.Coupe@vumc.nl)
P. Cuijpers (p.cuijpers@vu.nl)
M.L. Lee (mvanderlee@hdi.nl)
B.I. Lissenberg-Witte (b.lissenberg@vumc.nl)
B. Wijnen (bwijnen@trimbos.nl)
J.B. Prins (judith.prins@radboudumc.nl)
Irma Verdonck-de Leeuw (im.verdonck@vumc.nl)

Version: 1 Date: 17 Oct 2019

Author’s response to reviews:

Executive editor, BMC Psychology

October 17, 2019
Dear Dr. Darren Byrne

Please find enclosed our revised manuscript entitled: “Prevalence of adjustment disorder among cancer patients, and the reach, effectiveness, cost-utility and budget impact of tailored psychological treatment: study protocol of a randomized controlled trial”, which we would like you to consider for publication in BMC Psychology.

As you noticed the participants will indeed be selected from the Netherlands Cancer Registry database which is not publicly available. The data is only available for research and can be requested from IKNL. All necessary permissions were obtained to access and use the data and who gave this permission. Therefor we added the requested statement in the method section of the manuscript.

We hope that the manuscript meets with your approval and thank you in advance for your consideration.

Looking forward to your response,

On behalf of all authors,

Sincerely,

Prof. dr. Irma M. Verdonck-de Leeuw
VU University Medical Center, Department of Otolaryngology/Head &amp; Neck Surgery
P.O. Box 7057
1007 MB Amsterdam
Telephone: +31 20 4440931 / E-mail: im.verdonck@amsterdamumc.nl